Accelerator Status Report
Run Schedule Jan. 14, 2009 to March 12, 2009

· Injector energy 65.64 MeV
· Linac Setup 583. 5 MeV

· Pass setup: 5/5/4
· The accelerator is stable.

· Arc3A, MAG9R01 and MAM3S03 were not stable. The problems were addressed.
· Check energy match during a pass change,
· Leave orbit locks off when current is low in the machine (2uA?)
· The accelerator has been stable.

· The trim card for MFA0I03 was replaced, which made Hall A Compton rates more stable.

.

Hall A
E06-014    1.23GeV     1 Pass

Measurement of Single Target-Spin Asymmetry
· Compton rates are not stable. The Compton detector may not be reliable. Hall A beam is stable. The hall will report the status of its detector to us. 

· Hall A raster should be initialized during a pass change.

· Be careful with Hall A Moller setup. Reduce the ep ion chamber trips by adjusting 1H01Y up by 0.5mm.

· Hall A Moller procedure was revised: turn off the Compton orbit lock.

· Moller quad MQO1H03A has problems when restored from a Moller run.

Hall B
Eg1-DVCS    5.9 GeV    5pass

· Hall B beam positions are not stable some times. Monitor them using the StripTool.
· H2 Sniffer is masked.
· The hall is down. They finished their experiment.

· The hall is planning to come back on Thursday swing shift (2/5).
· The COO is available in the MCC.

Hall C

E07-003-SANE     5.9 GeV    5 Pass 
· Hall C is at pass 5 now.

· Hall C beam is fine.
· Hall C had a lot of downtime due to its target related problems.

· The target coil FSD read back is not stable. We can adjust the FSD window to clear the fault.

